

### Crossover localisation is regulated by the neddylation posttranslational regulatory pathway

Marina Tagliaro Jahns, Daniel Vezon, Aurélie Chambon, Lucie Pereira, Matthieu Falque, Olivier Martin, Liudmila Chelysheva, Mathilde Grelon

### ▶ To cite this version:

Marina Tagliaro Jahns, Daniel Vezon, Aurélie Chambon, Lucie Pereira, Matthieu Falque, et al.. Crossover localisation is regulated by the neddylation posttranslational regulatory pathway. PLoS Biology, 2014, 12 (8), 10.1371/journal.pbio.1001930. hal-01204105

### HAL Id: hal-01204105 https://hal.science/hal-01204105

Submitted on 27 May 2020

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

### Crossover Localisation Is Regulated by the Neddylation Posttranslational Regulatory Pathway



# Marina Tagliaro Jahns<sup>1,2</sup>, Daniel Vezon<sup>1,2</sup>, Aurélie Chambon<sup>1,2</sup>, Lucie Pereira<sup>1,2</sup>, Matthieu Falque<sup>3</sup>, Olivier C. Martin<sup>3</sup>, Liudmila Chelysheva<sup>1,2</sup>, Mathilde Grelon<sup>1,2\*</sup>

1 INRA, Institut Jean-Pierre Bourgin, UMR 1318, ERL CNRS 3559, Saclay Plant Sciences, RD10, Versailles, France, 2 AgroParisTech, Institut Jean-Pierre Bourgin, UMR 1318, ERL CNRS 3559, Saclay Plant Sciences, RD10, Versailles, France, 3 Institut National de la Recherche Agronomique, Unité Mixte de Recherche de Génétique Végétale, Université Paris-Sud, Gif-sur-Yvette, France

#### Abstract

Crossovers (COs) are at the origin of genetic variability, occurring across successive generations, and they are also essential for the correct segregation of chromosomes during meiosis. Their number and position are precisely controlled, however the mechanisms underlying these controls are poorly understood. Neddylation/rubylation is a regulatory pathway of posttranslational protein modification that is required for numerous cellular processes in eukaryotes, but has not yet been linked to homologous recombination. In a screen for meiotic recombination-defective mutants, we identified several *axr1* alleles, disrupting the gene encoding the E1 enzyme of the neddylation complex in *Arabidopsis*. Using genetic and cytological approaches we found that *axr1* mutants are characterised by a shortage in bivalent formation correlated with strong synapsis defects. We determined that the bivalent shortage in *axr1* is not due to a general decrease in CO formation but rather due to a mislocalisation of class I COs. In *axr1*, as in wild type, COs are still under the control of the ZMM group of proteins. However, in contrast to wild type, they tend to cluster together and no longer follow the obligatory CO rule. Lastly, we showed that this deregulation of CO localisation is likely to be mediated by the activity of a cullin 4 RING ligase, known to be involved in DNA damage sensing during somatic DNA repair and mouse spermatogenesis. In conclusion, we provide evidence that the neddylation/rubylation pathway of protein modification is a key regulator of meiotic recombination. We propose that rather than regulating the number of recombination events, this pathway regulates their localisation, through the activation of cullin 4 RING ligase complexes. Possible targets for these ligases are discussed.

Citation: Jahns MT, Vezon D, Chambon A, Pereira L, Falque M, et al. (2014) Crossover Localisation Is Regulated by the Neddylation Posttranslational Regulatory Pathway. PLoS Biol 12(8): e1001930. doi:10.1371/journal.pbio.1001930

Academic Editor: Michael Lichten, National Cancer Institute, United States of America

Received January 14, 2014; Accepted July 3, 2014; Published August 12, 2014

**Copyright:** © 2014 Jahns et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This work was funded by the ANR (Agence Nationale pour la Recherche). MTJ was funded by an INRA PhD fellowship (Contrat Jeune Scientifique, CJS). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

**Abbreviations:** CO, crossover; CRL, Cullin RING ligase; D-loop, displacement loop; DSB, double-strand break; HR, homologous recombination; IR, interference ratio; MCN, minimum chiasma number; NCO, gene conversion not associated with CO; NPD, nonparental ditype; PD, parental ditype; SDSA, synthesis-dependent strand annealing; ssDNA, single stranded DNA; T, tetratype.

\* Email: mathilde.grelon@versailles.inra.fr

#### Introduction

Meiosis is a modified cell cycle where two rounds of chromosome segregation follow a single S phase, resulting in the production of haploid gametes. Recombination is a key step in meiosis I, as it results in genetic crossover (CO) formation, which establishes physical links between the homologues cytologically visible as chiasmata [1,2]. In most species, each chromosome pair has at least one CO (referred to as the obligatory CO), which is required to hold the homologues together during the first meiotic division, ensuring their correct segregation. In most organisms, homologues that lack a CO often segregate improperly, leading to the formation of aneuploid gametes [3]. Meiotic recombination can also lead to gene conversion not associated with COs (NCOs) [4].

Meiotic recombination is initiated by the induction of DNA double-strand breaks (DSBs) catalysed by SPO11 [5]. DSBs are then resected by exonucleases to generate 3' single-stranded DNA molecules (ssDNA). In the subsequent step, RecA homologues RAD51 and DMC1 assemble on the ssDNA to form nucleoprotein

filaments. These filaments search for homologous sequences and trigger single-strand invasions [6] to generate displacement loop (D-loop) recombination intermediates [7]. Depending on the way these D-loop intermediates are processed, different recombination products can be formed. For example, capture of the second DSB end leads to the formation of a double Holliday junction that can be resolved to generate either a non-CO (NCO) or a CO [8–10]. Alternatively, NCOs can also be formed when a single strand end is displaced after priming a limited amount of DNA synthesis, annealing with the other DSB end in a process called synthesis-dependent strand annealing (SDSA) [11].

In most organisms, when multiple COs occur on the same chromosome, they are distributed nonrandomly: One CO prevents other COs from occurring close by, in a distancedependent manner. This phenomenon results in COs being more evenly spaced along chromosomes than would be expected if they occurred randomly. The term used to describe this phenomenon is CO interference [12,13]. In budding yeast, two kinds of COs are known to coexist: class I COs, which are interference-sensitive COs and whose formation depends on the ZMM proteins (Zip1,

#### **Author Summary**

During meiosis, two successive chromosomal divisions follow a single S phase, resulting in the formation of four haploid cells, each with half of the parental genetic material. This reduction in chromosome number occurs during the first meiotic division, when homologous chromosomes (paternal and maternal) are separated from each other. For this to happen, homologous chromosomes associate in structures called bivalents, where each chromosome is linked to its homologue by a point of contact known as chiasmata. These chiasmata reflect the formation of crossovers (COs), one of the manifestations of the exchange of genetic material occurring during homologous recombination. CO number varies little at around two per chromosome pair, and they tend to be evenly spaced on chromosomes. Thus, CO number and distribution are very tightly controlled. However, the mechanisms underlying these controls are very poorly understood. In this study, we identified a regulatory pathway of meiotic recombination. We show that this pathway does not regulate the amount of recombination events per se, but instead controls their localisation, as when it is defective, CO events cluster together in a few regions of the genome, leading to bivalent shortage and progeny aneuploidy with incorrect numbers of chromosomes. This regulatory pathway is a posttranslational protein modification system called neddylation (or rubylation in plants), known to be required for numerous cellular processes in eukaryotes. We identify an enzyme of the neddylation complex as a major regulator of meiotic recombination in Arabidopsis and show that this process may be also conserved in mammals.

Zip2, Zip3, Zip4, Msh4, Msh5 and Mer3) in addition to Mlh1 and Mlh3, and class II COs, which are not subject to interference and depend on Mus81 and Eme1/Mms4 [10]. *Arabidopsis thaliana*, like yeast and mammals, has two recombination pathways: one that exhibits CO interference and another one that does not [14–20]. In *A. thaliana*, disruption of genes acting in the interference-sensitive pathway causes a loss of approximately 85% of COs [21]. In addition, there is evidence that the *MUS81* gene accounts for some, but not all, of the 15% *MSH4*-independent COs, suggesting that MUS81 is involved in a secondary subset of meiotic COs that are interference insensitive [14,22]. Very little information is available on the mechanisms controlling interference and the number and distribution of COs during meiosis in general [23,24].

Eukaryotes possess a highly conserved mechanism to control protein degradation mediated by the action of the ubiquitin (Ub) proteasome system (UPS) [25]. In this system, E3 Ub ligases are required to ubiquitylate specific protein targets. Cullin RING ligases (CRLs) are the largest class of E3 ligases. Several mechanisms control CRL activity: It can be activated by covalent attachment of the Ub-like protein NEDD8/RUB (a process called neddylation or rubylation) [26,27] or inhibited by the COP9 signalosome-directed deneddylation [28]. Neddylation/rubylation has been shown to play a crucial role in processes such as morphogenesis in mice [29], cell division in budding yeast [30], embryogenesis in *C. elegans* [31], meiosis to mitosis transition in *C. elegans* [32], and response to various plant hormones [33,34] including auxin [35–38]. However, neddylation/rubylation had not been connected to homologous recombination (HR).

Cullin RING Ligase 4 (CRL4) is associated with DNA repair in plants and humans; the DDB1-CUL4A<sup>DDB2</sup> E3 ligase initiates nucleotide excision repair (NER) by recognizing damaged

chromatin with concomitant ubiquitylation of core histones at the lesion site [39–41]. Additionally, CUL4A plays a role in meiotic recombination and spermatogenesis in mice [42,43]. Inactivation of *cul4a* affected male fertility, with increased death of pachytene/diplotene cells and defects in MLH1 dissociation from the SCs.

Here we show that the E1 enzyme of the neddylation complex, AXR1, is a major regulator of meiotic recombination in *Arabidopsis*. In *axr1* mutants, the average number of meiotic COs is unchanged; they are still under the control of the ZMM proteins, but they tend to cluster together and no longer follow the obligatory CO rule. We were able to show that this recombination defect is correlated with strong synapsis defects. In addition, we found that this deregulation of CO localisation is likely mediated by a CRL4.

#### Results

#### The axr1 Mutants Are Meiosis-Defective

In the process of screening A. thaliana T-DNA (Agrobacterium tumefaciens transferred DNA) insertional lines for meiotic defects, we isolated three mutants [EGS344, EIC174, and EVM8 (Ws-4 strain); Figure 1 and Figure S1] allelic for disruption in At1g05180, the AXR1 gene, previously shown to encode the E1 enzyme of the Arabidopsis neddylation complex [44]. Another insertion line in At1g05180 available in the public collection (http://signal.salk.edu/) Sail\_904E06 (N877898, Col-0 strain) and the historical axr1 allele (axr1-12/N3076, Col-0 ecotype [44], with a single nucleotide substitution in exon 11 of At1g05180) were also included in this study (Figure 1).

The mutant plants all show the same vegetative phenotypes as previously described for axr1 mutants: They are dwarfed, excessively branched, with small rosettes and crinkled leaves (shown for N877898 in Figure 2A–B and in Figure S2 for the

500bp



Figure 1. The AXR1 gene and axr1 mutations. The arrow indicates the orientation of the open reading frame. Exons are shown as boxes (pink, UTR; black, CDS). In the EGS344 mutant, a large deletion associated with an insertion of the exogenous Aarobaterium Ti plasmid disrupts the AXR1 gene from nucleotide 91 (40 bp 5' to the ATG). In the EVM8 mutant, an in-frame deletion of 312 bp between exons 3 and 4 generates a 20 aa truncated protein. In EIC174, a single nucleotide insertion (A) in exon 6 (position 1364 of the genomic sequence, corresponding to nt 688 in the cDNA) leads to a premature stop codon (a 222 aa protein is produced instead of the 540 aa protein in wild type). In axr1-12, corresponding to the N3076 line, a single C-T nucleotide substitution at position 1295 of the cDNA leads to a premature stop codon (415 aa instead of 540), as described by Leyser et al. [44]. In N877898, corresponding to the Sail\_904E06 line, a T-DNA insertion occurred in intron 11. References used for this figure are Tair accession 4010763662 for the genomic sequence and Tair accession 4010730885 for the cDNA sequence.

doi:10.1371/journal.pbio.1001930.g001



**Figure 2.** *axr1* **developmental defects.** Five- (A) or nine- (B) weekold wild-type (wt) or *axr1* (N877898) mutant plants. *axr1* mutants are dwarfed, strongly branched, and have short siliques. Alexander staining (C–D) reveals round pollen grains, with a red cytoplasm reflecting viable male gametophytes in wild type (C), whereas *axr1* (N877898) anthers (D) contain a mixture of viable and dead (uncoloured, arrows) pollen grains. DIC microscopy of male meiosis products (E and F) reveals tetrads of microspores in wild-type (E) and unbalanced tetrads or polyads in *axr1* (F, N877898).

doi:10.1371/journal.pbio.1001930.g002

other alleles) [45,46]. They also have small flowers and short fruits, indicating fertility defects (Figure S2).

We examined the reproductive development of these mutants and found that all alleles showed a high level of male and female gametophyte abortion [shown for N8779898 male gametophytes (pollen grains) in Figure 2D]. In plants, male gametogenesis occurs in the anthers where groups of meiocytes undergo meiosis synchronously, each producing four haploid cells (called microspores). The four products of each meiosis remain temporally encased in a common callose wall, forming tetrads of microspores that can be visualised after tissue clearing (Figure 2E). Each microspore is then released from its tetrad and continues to develop into a mature pollen grain (the male gametophyte) containing the male gametes. Study of the early stages of pollen development in *axr1* revealed the presence of abnormal meiotic products. Instead of the regular tetrahedral structure observed in the wild-type, asymmetric tetrads (containing four daughter cells of unequal size) or "polyads" (containing more than four products) were observed (Figure 2F), suggesting that the meiotic program is disrupted in these mutants.

To confirm that the reduced fertility was caused by a defect in meiosis, we investigated male meiosis via chromosome spreading and DAPI (4',6-diamidino-2-phenylindole) staining (Figure 3). During wild-type meiotic prophase I (Figure 3A–D), DNA fibres of each sister chromatid are organised as chromatin loops connected to a common protein axis (the axial element [AE]) [47]. When chromosomes start to condense at leptotene, they become visible as threads (Figure 3A). At this stage, meiotic recombination is initiated by the formation of a large number of DNA DSBs (not shown). HR repairs these breaks concomitantly with the progression of synapsis, the close association of the homologous chromosome axes through the polymerisation of the central element (CE) of the synaptonemal complex (SC). Synapsis begins at zygotene (not shown) and is complete by pachytene, when complete alignment of homologous pairs can be detected in DAPI-stained chromosomes (Figure 3B). DNA repair and recombination are thought to be achieved during pachytene, yielding at least one CO per homologous chromosome pair. At diplotene (Figure 3C), when the CE of the SC is depolymerised, the homologous chromosomes are therefore connected to each other by COs in which chromatids from homologous chromosomes have been exchanged. These connections between homologous chromosomes become apparent only at diakinesis (Figure 3D, arrows), when chromosomes are sufficiently condensed. At this stage in Arabidopsis, chiasmata (the cytological manifestations of COs) cannot be scored precisely, but chiasma-carrying chromosome arms can sometimes be identified based on bivalent appearance (see Figure 3D, arrows). Next, condensation proceeds and, at metaphase I, the five Arabidopsis bivalents are easily distinguishable, aligned on the metaphase plate (Figure 3E). During anaphase I, sister chromatid cohesion is released from chromosome arms, allowing homologous chromosomes to segregate to the two opposite cellular poles (Figure 3F). The second meiotic division then separates the sister chromatids, generating four pools of five chromosomes (Figure 3G and 3H), which gives rise to the tetrads of four spores (Figure 2E).

In A. thaliana axr1 mutants, the leptotene and zygotene stages appeared similar to those in the wild type. However, no pachytene cells were identified in the 457 meiocytes analysed, in contrast to wild type, where this stage is present in approximately 35% of the cells (n = 334). Instead, we observed pachytene-like stages, with only partial chromosome alignment (Figure 3J). This suggests that axr1 is defective in synapsis. Diplotene cells were indistinguishable from those in the wild type (Figure 3K). Then, chromosome condensation could be followed until metaphase I, although diakinesis stages were rarely observed (1% of all stage cells, n = 457for N877898, 12% in wt, n = 334) (Figure 3L).

At wild-type metaphase I, the five typical *Arabidopsis* bivalents could be observed aligned on the metaphase plate (Figure 3E). Each bivalent was composed of two homologous chromosomes connected by chiasmata either on one chromosome arm (rod bivalent, Figure 3E#) or on both pairs of chromosome arms (ring bivalent, Figure 3E\*). Chiasma numbers could therefore be estimated based on the bivalent structure. However, because multiple COs on a single arm cannot be cytologically differentiated from single COs, these estimates only correspond to a minimum chiasma number (MCN; Figure 4, Table S1).



**Figure 3.** *axr1* **mutants show normal meiotic progression but reduced bivalent formation at metaphase I.** DAPI staining of meiotic chromosomes in wild type (A–H) and *axr1* (N877898, I–P). At the onset of meiotic prophase I (A and I), chromosomes can be identified. Chromosome alignment and synapsis then proceeds, leading eventually to the pachytene stage in wild type (B), where homologous chromosomes are synapsed along their entire length. This association can be observed in *axr1* (J, enlarged regions) but remains partial. Then, the SC disappears at diplotene (C and K), condensation proceeds, and bivalents can be identified in wild type at diakinesis (D), but this stage is rarely observed in *axr1* (L). At metaphase I, the five *Arabidopsis* bivalents can be identified in wild type (E), segregating at anaphase I (F). In *axr1*, a mixture of bivalents and univalents are observed (M), leading to subsequent improper segregation at anaphase I (N). Sister chromatids segregate at meiosis II (G and O), leading to balanced tetrads in wild type (H), unbalanced tetrads (not shown) or polyads in *axr1* (P). At metaphase I, univalents (u) can be distinguished from ring bivalents (where a chiasma occurred in each of the two chromosome arms, \*) and from rod bivalents (where only one chromosome arm shows a chiasma, #). Arrows in (D) indicate some of the chiasma-containing arms. Bar, 10  $\mu$ m.

In *axr1* mutants, we observed reduced bivalent formation, and instead of five bivalents, a mixture of bivalents and univalents could be identified (Figure 3M). The reduction in bivalent formation resulted in chromosome mis-segregation during subsequent anaphase I (Figure 3N), whereas the second meiotic division separated sister chromatids (Figure 3O), giving rise to a variable number of daughter cells containing aberrant numbers of chromosomes (Figure 3P).

We quantified the decrease in bivalent formation as well as the MCN at metaphase I from all axr1 mutants and their respective wild-type accessions (Figure 4, Table S1). On average, axr1 mutants had 78% of the wild-type number of bivalents for the Col-0 background and 52% for the Ws background. In terms of the chiasma number, axr1 mutants displayed a residual level of 56% and 41% of the wild-type levels for Col-0 and Ws strains, respectively (Figure 4). Within a single ecotype (Col-0 or Ws), all alleles were statistically different from the wild type but not different from each other. Finally, when the partitioning of the residual chiasmata in axr1 was analysed, we observed that a large

proportion of metaphase I cells showed both ring bivalents (at least two chiasmata) together with univalents (no chiasma) (42% of the N877898 cells, n = 47), showing that in *axr1*, the obligatory CO is lost.

To further analyse the bivalent shortage observed in axr1, we used fluorescence *in situ* hybridization (FISH) analyses on PMCs. Metaphase I chromosomes were labelled with probes for the 45S and 5S rDNA repeats, allowing specific identification of chromosomes 1, 2, and 4 (Figure 5). Chromosomes 3 and 5 could not be discriminated from each other with these probes and were pooled. First, we observed that in axr1 as in wild type, bivalents were always formed between homologous chromosomes (n = 147) bivalents for axr1, n = 165 for wt). Then, we considered each bivalent individually and determined which pair of chromosomes was involved in its formation. As shown in Figure 5D, in axr1, as in the wild type, each pair of chromosomes was equally involved in bivalent formation, showing that the decrease in bivalent formation observed in axr1 affected all chromosomes in the same way.



**Figure 4. COs in** *axr1* **are largely ZMM-dependent.** For each *axr1* allele and for their respective wild-type strains (Ws-4 for EGS344, EIC174, EVM8, and Col-0 for N8777898 and N3076), and for a combination of multiple mutants, the level of bivalent formation as well as the MCN per cell were measured. For multiple mutant analyses, the N877898 allele was used in a Col-0 background and EGS344 in a Ws-4 background. The complete dataset can be found in Table S1. doi:10.1371/journal.pbio.1001930.q004

#### axr1 COs Are ZMM-Dependent

In wild-type *Arabidopsis*, the majority of COs (85%–90%, depending on the genetic background Col-0 versus Ws-4) depend on the ZMM proteins (MSH4, MSH5, MER3, ZIP4, SHOC1/ZIP2, HEI10, and PTD) as well as on MLH1 and MLH3 [21,48],



**Figure 5. Bivalent shortage has a similar effect on each pair of chromosomes.** Fluorescent in situ hybridisation (FISH) on metaphase I cells was performed with probes directed against the 45S (green) and the 5S (red) rDNA, which allow the identification of chromosomes 1 (unlabelled), 2 (green labelled), and 4 (green and red labelled), whereas chromosomes 3 and 5 cannot be distinguished (red labelled). In wild type, each chromosome pair represents 20% of the total number of bivalents (A and D, centre circle, in light, n = 21 cells). In *axr1* (B and D, N877898 allele, external circle, n = 28), the proportion of each bivalent pair is the same as in wild type. Bar = 5  $\mu$ m. doi:10.1371/journal.pbio.1001930.g005

whereas MUS81 is responsible for 10%–15% of the remaining COs [14,22].

We measured bivalent formation frequencies and the chiasma frequencies in various genetic combinations compared to the single axrI mutant (Figure 4, Table S1). For all the *zmmaxrI* double mutants (except *mer3axrI*) and regardless of strain (Col-0 versus Ws-4), the level of bivalent formation was reduced by more than 95% with hardly any bivalents observed (from 0.13 to 0.18 bivalent per cell; Table S1), showing that almost all the COs in axrI are ZMM-dependent.

We also analysed the bivalent frequency in the axr1mus81 double mutant, which was the same as for the axr1 single mutant  $(3.77\pm1.03 \text{ against } 3.75\pm1.12; p=0.9)$  (Figure 4). We then quantified bivalent frequency in the axr1msh5mus81 triple mutant and observed, as expected, a dramatic decrease in bivalent formation compared to axr1mus81 (Figure 4). No difference could be detected between the axr1msh5mus81 triple mutant and the axr1msh5 double mutant (p=0.2). These results show that CO formation in axr1 mutants is almost exclusively dependent on ZMM proteins, whereas the MUS81 pathway plays only a limited role, if any.

#### Class I COs Are Mislocalised in the axr1 Mutant

To further analyse recombination events in *axr1*, we immunolabelled chromosomes with antibodies directed against HEI10 and MLH1, two markers of class I COs in Arabidopsis [48,49]. MLH1 foci can be seen from late pachytene to diakinesis [49], whereas HEI10 is first loaded early during prophase on a large number of sites forming foci of different sizes on chromosomes. A limited number of these foci then remain (Figure 6A and B) at sites that correspond to class I COs where they co-localise with MLH1 until the end of prophase [48]. We therefore counted HEI10 and MLH1 foci in late pachytene and diplotene cells in wild type and axr1. Surprisingly, the average foci number per cell was not different between wild type and axr1, for either HEI10  $(8.30\pm0.29, n=54 \text{ and } 7.49\pm0.40, n=84, p=0.15)$  or MLH1  $(8.61 \pm 0.29, n = 33 \text{ and } 7.58 \pm 0.54, n = 91, \text{ respectively},$ p = 0.263). In addition, we confirmed that these foci localise to chiasma-containing arms at diakinesis (Figure 6E and F and Figure S3), showing that they are likely to mark CO sites in axr1 as



Figure 6. The average number of class I COs is similar in wild type and *axr1*. HEI10 or MLH1 was immunolocalised on acetic acid spread chromosomes from wild-type (A, B, and E, Col-0) or *axr1* (C, D, and F) meiocytes from late pachytene to diakinesis. In *axr1* (N877898 allele), the average number of HEI10 or MLH1 foci per cell is similar to that in wild type (G). Bar =  $5 \mu$ M. doi:10.1371/journal.pbio.1001930.q006

in wild type [49]. We also observed that there was higher variability in the numbers of HEI10 and MLH1 foci in axr1 than in wild type (Figure 6G), with the coefficient of variation (standard deviation divided by the mean) varying from 26% (HEI10, wt) to 50% (HEI10, axr1) or from 19% (MLH1, wt) to 68% (MLH1, axr1).

Another striking feature of *axr1* was the frequent occurrence at the pachytene-like and diplotene stages of portions of paired chromosome axes where adjacent HEI10 and MLH1 foci could be seen (Figure 6C, D, arrows and Figure 7A, arrows). Forty-seven percent (HEI10, n = 60) or 53% (MLH1, n = 66) of the cells had at least two foci localised on the same portion of a chromosome axis, whereas in wild type, this scenario occurred only in 7% (HEI10, n = 57) or 3% of the cells (MLH1, n = 39) (Figure 7B). In addition, although we never observed more than two adjacent foci in wild type, we observed 22% (HEI10) and 13% (MLH1) of the cells with more than two adjacent foci, with a maximum of five adjacent HEI10 foci observed in *axr1* (Figure 7B). Therefore, although the average level of class I COs is the same in *axr1* and in wild type (Figure 6G), these class I COs tend to cluster together in at least 50% of the *axr1* cells.

We then estimated the scale at which this clustering arises. The distance between clustered foci was measured and compared to the total length of chromosome axis. The distance between two adjacent foci was on average 1/400 of the total axis length of a cell, ranging from 1/1600 of the genome to a maximum of 1/90 of the genome (Figure S4A). Extrapolated in DNA distance, with the additional assumption that genome condensation is homogeneous, the distance between two adjacent foci in a cluster is therefore expected to vary from 150 kb to 3,000 kb, with an average of 625 kb. We also observed that the distance between two adjacent foci does not vary significantly in clusters with exactly two foci compared with clusters with more than two foci. As a consequence, cluster size increases proportionally with the number of foci present in the cluster (Figure S4B). The size of the clusters was on average 1/200 of the genome for HEI10 foci (n = 14, 1, 200kb) and 1/300 for MLH1 (n = 21, 800 kb).

Finally, we examined whether the clustered foci displayed interference, as might be expected for class I COs. We thus considered the hypothesis  $H_0$  that the foci in clusters are not subject to interference. The test was based on the distribution of distances between adjacent foci, specifically using the coefficient of variation for the statistical test and comparing to  $10^5$  simulations under  $H_0$  (see Materials and Methods). For the clusters of three or more foci (Table S2), we rejected the  $H_0$  hypothesis of no interference for MLH1 foci (p = 0.0024 based on seven clusters), for HEI10 foci (p = 0.0028 based on six clusters), and when pooling the MLH1 and HEI10 data ( $p = 2.4 \times 10^{-5}$  based on 13 clusters). Specifically, inside clusters, MLH1 and HEI10 foci are more evenly distributed than at random, showing that COs within clusters still interfere.

Taken together, these results show that the shortage in bivalent formation observed in *axr1* mutants is not due to a general decrease in CO formation but rather to a mislocalisation of class I COs that tend to cluster together.

#### Measurement of Recombination Rates in axr1 Mutants

The level of genetic recombination on several chromosomal intervals was measured using the Fluorescent-Tagged Lines (FTL) tool developed by Copenhaver et al. [50]. The FTL system is a visual assay based on segregation of genetically linked fluorescent proteins expressed in the pollen grains of the quartet mutant (*qrt1*), in which the pollen grains remain attached as tetrads. With these lines, a large number of meiotic products can be visually scored and then a subset of multiple CO events can be identified (two-, three-, and four-strand double COs in adjacent intervals and four-strand double COs within a single interval) ([50] and Table S3B). Six different intervals were used, either on chromosome 3 (I3b and I3c) or 5 (I5a, I5b, I5c, and I5d), with sizes ranging from 1,200 to 4,900 kb (Table S3A).



**Figure 7. Class I COs tend to cluster in** *axr1*. (A) Examples of adjacent HEI10 or MLH1 foci in wild-type (Wt, Col-0) and in *axr1* (N877898 allele) acetic acid spread meiotic chromosomes. (B) Proportion of pachytene and diplotene cells where adjacent foci were observed on the same chromosome axis pair (Wt, wild type; *axr1*, N877898 allele) (0, no evidence of adjacent foci; 2, two adjacent foci, etc.). Some of these situations are indicated by arrows in panel A. Bar =  $5 \mu M$ . doi:10.1371/journal.pbio.1001930.g007

We first measured recombination rates for each interval using the standard Perkins genetic mapping equation [51]. As shown in Table 1, recombination rates in axr1 vary differently depending on the interval tested, from 70% to 180% of the wild-type level. On average, axr1 shows an increase in recombination, but these data should be taken with caution, as recombination measurements rely only on a subset of tetrads (the viable tetrads). Out of the six intervals considered, intervals located close to the telomeres (I3b and I5b) showed the most significant increase in recombination, whereas proximal intervals appeared less affected. This could indicate that the level of recombination is affected differently according to the location on the chromosomes, although additional data will be required to determine if telomere proximity increases CO frequency in the mutant.

We then used the FTL data to estimate interference between COs occurring in adjacent intervals (Table 2 and Table S4). We calculated the Interference Ratio (IR) as defined by Malkova et al. [18], which compares the genetic length of one interval with and without the presence of a simultaneous event in the neighbouring

interval. When the occurrence of a CO in one interval reduces the probability of a CO occurring in the adjacent interval, the IR is less than 1, indicating (positive) CO interference. When COs in the two adjacent intervals are independent of each other, the IR is 1, and if the presence of one CO in an interval increases the probability of an additional CO in the adjacent interval, the IR is greater than 1, indicating negative interference. As shown in Table 2, all wild-type IRs were less than 1, in agreement with the presence of CO interference. For axr1, however, all IRs increased dramatically and were statistically significantly different to wild type (p < 0.0001, Table 2 and Table S4). In addition, all axr1 IR values were greater than 1, although only one pair of intervals tested was significantly different from 1 (I5a I5b, first data set, IR = 1.63,  $p = 4 \times 10^{-3}$ ). Therefore, in *axr1*, adjacent COs appear to occur more frequently than in wild type, which is in agreement with the previously observed clustering of class I COs scored cytologically (Figure 7). The cytologically observed clustering is occurring at a very small scale, namely a few hundred kb (on average 1,200 kb for HEI10 foci and 800 kb for MLH1 foci, see Table 1. Recombination rates and interinterval interference.

Туре	Intervals	N° of Tetrads <sup>c</sup>	d (CM) <sup>d</sup>	d ratio ( <i>axr1</i> /wt) <sup>e</sup>	NPD Ratio <sup>f</sup>	$p (NPDr = 1)^{g}$
Wt	l5a <sup>a</sup>	3,118	24.2	-	0.27	<10 <sup>-12</sup>
	l5b <sup>a</sup>	3,118	15.5	—	0.37	<10 <sup>-3</sup>
	I5a <sup>b</sup>	1,899	27.9	—	0.27	<10 <sup>-11</sup>
	I5b <sup>b</sup>	1,899	16.7	—	0.27	<10 <sup>-3</sup>
	I5d	8,947	8.6	—	0.21	10 <sup>-5</sup>
	l5c	8,947	8.9	—	0.35	<10 <sup>-3</sup>
	l3c	10,245	4.8	—	0.43	0.1
	l3b	10,245	17.0	_	0.30	<10 <sup>-18</sup>
axr1	l5aª	1,263	17.0	0.7**	2.69	<10 <sup>-7</sup>
	l5b <sup>a</sup>	1,263	23.7	1.5**	1.63	<10 <sup>-2</sup>
	I5a <sup>b</sup>	1,107	24.4	0.9*	1.47	0.05
	I5b <sup>b</sup>	1,107	29.9	1.8**	1.18	0.05
	l5d	2,274	9.1	1.1	0.82	0.9
	l5c	2,274	8.8	1.0	0.74	0.7
	l3c	1,499	4.9	1.0	1.24	0.9
	I3b	1,499	20.9	1.2**	1.24	0.4

<sup>a, b</sup>Correspond to the data obtained for two independent experiments for intervals I5a and I5b.

<sup>c</sup>Only four-spore viable tetrads were considered. They correspond to 97% of the tetrads (n = 3,756) in wild type and 10% of the tetrads (n = 5,973) for axr1.

<sup>d</sup>Map distances were calculated using the Perkins genetic map equation [51] using raw data from Table S3B.

<sup>e</sup>Genetic distance ratio between *axr1* and wild type. It compares recombination rates between the two genotypes. Asterisks indicate significant differences between mutant and wild type (\* p<0.05; \*\* p<0.01).

<sup>f</sup>Ratio between the observed number of double COs (based on NPD tetrad frequency) to the expected number of double COs under the hypothesis of no interference (see Table S5). The NPDr gives the strength of interference within the considered interval (no interference if the NPDr is equal to 1, absolute interference if the NPDr is equal to 0, negative interference if the NPDr is above 1).

<sup>9</sup>The *p* values indicate significant differences between IR and 1.

doi:10.1371/journal.pbio.1001930.t001

above), whereas in FTLs pairs of intervals correspond to more than 3,000 kb (I5cd, I3bc) and up to 7,500 kb (I5ab). Consequently, most of the clusters are expected to be present within a single interval and to only occasionally affect two adjacent intervals, which could explain why only one pair of intervals showed significant negative interference.

Double CO events within a single interval can be detected using the FTLs if the two COs involve four different chromatids (Table S3B) because they will generate nonparental ditype (NPD) tetrads [50]. Interference within single intervals can be estimated by comparing the observed number of double COs (NPD frequency) to the expected number of double COs under the hypothesis of no interference [52]. The ratio between these two numbers (NPDr) gives the strength of interference within the considered interval, even if an important proportion of multiple COs will be silent. We calculated NPDr for all intervals considered for wild type and *axr1* 

Туре	Adjacent Intervals	<i>N</i> ° of Tetrads <sup>c</sup>	IR <sup>d</sup>	$p(IR = IR wt)^{e}$	$p(IR=1)^{f}$	
Wt	I5a/I5b <sup>a</sup>	3,118	0.47	_	<10 <sup>-8</sup>	
	I5a/I5b <sup>b</sup>	1,899	0.52	_	<10 <sup>-8</sup>	
	I5d/I5c	8,947	0.36	_	<10 <sup>-8</sup>	
	I3c/I3b	10,245	0.28	_	<10 <sup>-8</sup>	
axr1	I5a/I5b <sup>a</sup>	1,263	1.63	<10-4	4×10 <sup>-3</sup>	
	I5a/I5b <sup>b</sup>	1,107	1.25	<10-4	0.09	
	I5d/I5c	2,274	1.26	<10-4	0.11	
	l3c/l3b	1,499	1.35	<10-4	0.10	

#### Table 2. Intra-interval interference.

<sup>a, b</sup>Correspond to the data obtained for two independent experiments for intervals I5a and I5b.

<sup>c</sup>Only four-spore viable tetrads were considered. They correspond to 97% of the tetrads (n = 3,756) in wild type and 10% (n = 5,973) for axr1.

<sup>d</sup>The IR compares the genetic size of the first interval when a CO occurs in the adjacent interval to the genetic size of the same interval when no CO occurs in the adjacent interval (Table S4A).

<sup>e</sup>The *p* values indicate significant differences between *axr1* IR and wild-type IR for a given pair of intervals.

<sup>f</sup>The *p* values indicate significant differences between IR and 1.

doi:10.1371/journal.pbio.1001930.t002

(Table 1 and Table S5). In wild type, the NPDr indicated strong interference (NPDr close to 0.3) within all the intervals (except for I3c, which is too small for statistically meaningful data, Tables S3A and S5). In *axr1*, however, the NPDr increased systematically (between 0.7 and 1.47) and was mostly greater than 1. For two intervals (I5a and I5b), the NPDr values of 2.69 and 1.63 were statistically significant (p<0.01), showing negative interference (Table 2).

Thus, genetic analyses allowed us to measure negative interference in several of the intervals tested, confirming the CO clustering observed in cytology.

#### Recombination Initiation Is Not Modified in axr1 Mutants

To verify whether the recombination defect in axr1 could be linked to a defect in recombination initiation, we used two methods to investigate DSB formation. We first introgressed the axr1 mutation into a rad51 mutant, defective for meiotic DSB repair. In this mutant, DSBs are formed but are then repaired abnormally, leading to significant chromosomal defects (such as chromosome bridges and chromosome fragmentation) during anaphase I (Figure S5A). These chromosomal defects persisted in axr1rad51, showing that DSBs are present in the axr1 mutant (Figure S5B). Second, we analysed the nuclear distribution of the DMC1 protein, a meiosis-specific recombinase that forms foci at recombination sites. The dynamics and number of AtDMC1 foci in axr1 (237±40, n=7) were indistinguishable from wild type  $(234\pm89, n=28)$  (t, p=0.9) (Figure S5). Thus, the meiotic defects observed in axr1 are not correlated with a decrease in the amount of recombination initiation events.

## Synapsis Is Strongly Defective in axr1 But Chromosome Axes Are Normal

During meiotic prophase, chromosomes are structured in the context of a protein axis (the AE), which is crucial for most meiotic events, including meiotic recombination and synapsis [53,54]. The meiotic chromosome axis is composed of specific AE proteins, such as ASY1 and cohesion proteins (REC8 and SCC3, [55,56]). In wild-type meiotic cells, cohesins are loaded as early as premeiotic G1, whereas ASY1 appears at leptotene first as foci, then as a linear signal throughout the entire chromosome length (Figure S6A), in a pattern similar to that of cohesins (Figure S6C, [56]). As shown in Figure S6, the signal observed in axr1 mutants cannot be differentiated from wild type, showing that no major alteration of the axis can be detected in axr1 mutants.

We then analysed the progression of synapsis by immunolocalisation of ZYP1, the A. thaliana CE component [57]. In wild type, ZYP1 appeared on chromosomes as foci that quickly elongated to yield a mixture of foci and short stretches of ZYP1 (Figure 8A,B, red signal and Figure S7). Synapsis then progressed until complete synapsis was reached, defining the pachytene stage (Figure 8C,D and Figure S7). In axr1, the early stages of synapsis could not be distinguished from wild type, showing a mix of foci and short ZYP1 stretches (Figure 8E,I and Figure S7). As meiosis progressed, ZYP1 elongation could be detected (Figure 8F-L and Figure S7), but full synapsis was never achieved (n = 66), confirming the synapsis defect detected after DAPI staining of meiocyte spreads (Figure 3). In addition, in approximately half of the cells, ZYP1 signals appeared strongly perturbed, uneven in thickness and forming dotted lines rather than a homogeneous continuous signal (Figure 8J or G and Figure S7). In some cases, only short and thick ZYP1 stretches were detected. These could correspond to ZYP1 poly-complexes rather than to CE polymerisation (Figure 8L and Figure S7).

#### CO maturation and Synapsis Can Be Uncoupled in axr1

To follow the progression of meiotic recombination events, we co-immunolocalised ZYP1 and HEI10, using a lipsol spreading protocol that has the advantage of allowing the simultaneous detection of these two proteins [58] but also the disadvantage of preventing examination of prophase after pachytene [59]. As mentioned above, HEI10 is detected as foci on meiotic chromosomes from leptotene to diakinesis, and its dynamics reflect the progression from early recombination intermediates to mature class I COs [48]. During leptotene and early zygotene, HEI10 forms numerous foci of variable size on chromatin (Figure 8A and Figure S8). Then, during synapsis initiation, bigger and brighter HEI10 foci appear, often co-localising with synapsed regions (Figure 8B and Figure S8). At this stage and later on, a combination of large and small foci are observed, forming "strings of HEI10 pearls" on ZYP1 stretches (Figure 8B,C and Figure S8B,C, arrows). At late pachytene, only a few bright HEI10 foci, corresponding to mature class I COs, are retained (Figure 8D and Figure S8). Nevertheless, during most of the pachytene stage, bright HEI10 foci are present, together with faint HEI10 signal marking the CE (Figure S8C,D). In axr1, the dynamics of HEI10 progression were the same as in wild type with HEI10 detected as multiple foci during early prophase stages (Figure 8E,I and Figure S8). Brighter foci then appeared as synapsis progressed, also forming a string of pearls on ZYP1 stretches (Figure 8J and Figure S8, arrows). A subset of very bright foci was retained at the later stages (Figure 8G,H,K,L and Figure S8). We noticed that at these late stages (based on the HEI10 pattern), the level of synapsis varied considerably from one cell to another. In addition, although these late HEI10 foci were always observed on ZYP1 stretches, the reverse was not true and ZYP1 stretches without late HEI10 signals were observed (see, for example, Figure 8H, where four late HEI10 foci are clustered on a single ZYP1 stretch, whereas many ZYP1 stretches are deprived of HEI10 foci). Therefore, it appears that class I CO clustering in *axr1* is correlated with strong synapsis defects, but cannot be explained by the limited extension of the SC.

### Meiotic Defects in the *axr1* Mutant Are Epistatic to Those of a Cullin 4 Mutant

Because neddylation is known to regulate the activity of CRLs, we investigated whether *axr1* meiotic defects are dependent on a specific CRL. In *A. thaliana* only four cullins are neddylated: cullin 1, cullin 3A, cullin 3B, and cullin 4 [33]. To identify possible AXR1 downstream players, we scored cullin-deficient lines for meiotic defects. Complete suppression of any of cullin functions (null *cul1* or *cul4* or the double *cul3a cul3b* mutants) is lethal, but various genetic backgrounds deficient in cullin activities are available

We first investigated meiosis of the auxin response defective *cul1* mutant alleles—*cul1–6* [60], axr6-2/N3818 [61], and axr6-3/eta1 [62]—and observed perfectly normal meiosis (not shown). Next, considering cullin 3 activity, we analysed the CUL3a/3b hypomorphic mutant [*cul3w* (*cul3a3cul3b1*)] described for its defects in various aspects of the ethylene biosynthesis pathway and root development [63]. *cul3w* plants also showed normal meiotic development of male meiocytes (not shown).

Finally, we analysed the *cul4-1* mutant in which a T-DNA is inserted occurred in the 12th exon of the gene, leading to aberrant *CUL4* mRNA expression, which varies depending on the developmental stage [64]. We observed significant male and female gametophyte abortion in *cul4-1* (shown for the male, compare Figure 9A to Figure 2C). Although in wild type only balanced tetrads of microspores were observed, asymmetric tetrads



**Figure 8. Synapsis is strongly perturbed in** *axr1*, **but HEI10 dynamics during early prophase are unchanged**. ZYP1 and HEI10 proteins were co-immunolocalised on lipsol-spread chromosomes from wild-type (A–D) and *axr1* (N877989 allele, E–L) meiotic cells. The overlay of both signals is shown here (ZYP1 in red, HEI10 in green), but single channels can be found in Figures S7 and S8. In wild type as in *axr1*, ZYP1 appears on chromosomes as foci that quickly elongate, yielding a mixture of foci and short stretches (A, B, E, F, and I). Synapsis then progresses until complete synapsis is reached in wild type, defining the pachytene stage (C–D). In *axr1*, ZYP1 elongation can be detected, but full synapsis was never achieved (G, H, and K). In *axr1*, the ZYP1 signal is often uneven in thickness or forms dotted lines rather than a homogeneous and continuous signal (J and G). In addition, in some cases, only short and thick ZYP1 stretches were detected which could correspond to ZYP1 polycomplexes (L). During early zygotene, in wild type as in *axr1*, HEI10 forms numerous foci of variable sizes on chromatin (A, E, and I). Then, although synapsis progresses, combinations of large and small foci are observed, forming "strings of pearls" on ZYP1 stretches (B, F, and J, arrows). As meiosis progresses, a few bigger and brighter HEI10 foci can be observed in wild type (D) and in *axr1* (G, H, K, and L), which generally co-exist with smaller and fainter HEI10 foci (C, D, G, H, and K). Whereas this latter HEI10 pattern is associated with complete synapsis in wild type (C–D), synapsis is only partial in *axr1* (G, H, K, and L). Bar = 2  $\mu$ m. doi:10.1371/journal.pbio.1001930.g008

and polyads were seen in cul4-1 mutants (compare Figure 9B to Figure 2E). Male meiosis was then investigated. The first stages of meiosis proceeded normally in cul4-1 mutants, however we observed metaphase I phenotypes reminiscent of the axr1 defects,

with a large proportion of cells showing a clear reduction in bivalent formation (Figure 9C). The MCN per meiotic cell in *cul4-1* ( $6\pm3.2$ , n=71) was significantly different from wild type ( $8.9\pm0.9$ , n=51, p<0.0001), and slightly different from *axr1* 



Figure 9. CULLIN4 is involved in meiotic recombination in the same pathway as *axr1*. In the *cul4-1* mutant, a mixture of viable (purple) and dead (arrow) pollen grains can be seen in the anthers after Alexander staining (A). This is correlated with the production of aberrant tetrads and polyads of microspores (B). DAPI staining of the meiotic chromosomes revealed a defect at metaphase I (C) in bivalent formation, which is quantified in (E). Bar =  $10 \ \mu$ M. doi:10.1371/journal.pbio.1001930.q009

 $(5.1\pm1.5, n = 74, p = 0.02)$ . Nevertheless, the number of MCN per cell in *cul4-1* was far more variable than in *axr1* (Figure 9E), due to an overrepresentation of cells with wild-type levels of chiasmata (Figure 9D,E). We then introgressed the *axr1* mutation (N877898) into *cul4-1* and found that the double mutant cannot be distinguished from the single *axr1* in terms of meiotic phenotype (not shown), the average level of MCN per cell ( $4.9\pm1.8, n = 98, p = 0.412$ ), and in terms of variability of the values (Figure 9E), showing that *axr1* is epistatic to *cul4-1*. Overall, our results suggest that AXR1 acts during meiotic recombination through the activation of a CRL4 complex.

#### Discussion

## AXR1 Controls the Localisation of Class I COs During Meiosis

We observed that in *axr1* mutants, meiotic nondisjunction is correlated with defects in bivalent formation. However, our results indicate that the general level of meiotic recombination in *axr1* is close to that of wild type, as we showed that COs are mostly under the control of the ZMM pathway and that their average number, revealed by MLH1 and HEI10 foci, is unchanged. Furthermore, these CO events show a completely aberrant distribution in *axr1*. First, cytogenetic data showed that clustered MLH1 or HEI10 foci are observed in approximately 50% of the meiocytes (Figure 7B).

Second, genetic data showed that adjacent COs in most tested intervals no longer display genetic interference, showing that CO distribution is abnormal in axr1. More strikingly, in several intervals, strong significant negative interference was detected, a genetic demonstration of CO clustering. We can therefore conclude that COs in axr1 tend to cluster together and that the observed shortage in bivalent formation is not due to a global decrease in meiotic recombination but rather mislocalisation of these events, resulting in a loss of the obligatory CO.

Very little information is available on the mechanisms that control CO distribution during meiosis. Nevertheless, it has been known for a long time that COs are not randomly distributed among chromosomes, as in most organisms, adjacent COs display interference and therefore tend to be evenly spaced within chromosomes [13]. In addition, the phenomenon of the "obligatory CO" (or CO assurance) ensures the formation of at least one CO per bivalent, whatever the total number of CO precursors per cell. The relationship between these two phenomena is still under debate [23], but recent modelling analyses suggested that the obligatory CO is a direct consequence of interference [65]. Numerous mutants with altered interference were described, but they nearly always also change CO rates, either because of increased MUS81-dependant COs [66-68] or because they are defective in the ZMM CO pathway (see, for example, [48,69]). One possible exception is the Saccharomyces cerevisiae pch2 mutant, for which two independent studies showed that CO interference is alleviated without changes to meiotic recombination rates, at least on the smallest yeast chromosome (III) [70,71]. Nevertheless, the generalisation of this observation to the whole genome seems unlikely [71].

To our knowledge, *axr1* is therefore the first mutant that specifically modifies the localisation of class I COs, changing interference among them and resulting in the loss of the obligatory CO, but without changing the global average number of CO events. Thus, AXR1 is a key regulator of meiotic recombination outcomes.

From our data, we can exclude that the meiotic defects observed in axr1 are due to a major decrease in DSB formation or to a drastic mislocalisation of these events (Figure S5). We have also shown that axr1 meiotic defects are not associated with major chromosome axis defects (Figure S6), but instead with major perturbations in the polymerisation of the SC CE (Figure 8).

The relationship between CO control and SC polymerisation is a long-standing question in the field of meiosis [72]. In yeast, SC polymerisation is not necessary for CO interference, as it occurs after CO patterns have been imposed [73,74]. In Caenorhabditis elegans, however, it was recently shown that the SC central region limits the formation of COs and imposes total interference [75]. This could also be the case in rice, where *zep1* mutants (ZEP1 being the rice CE ZIP1 homologue) show an increase in chiasma formation at diakinesis [76]. This suggests that, in plants as in C. elegans, SC polymerisation could be necessary to limit CO formation. However, in Arabidopsis, ZYP1 appears to be required to prevent non-HR rather than acting on homologous CO formation [57]. To further complicate our understanding of the relationship between polymerisation of the SC CE and CO controls, in yeast, SC polymerisation requires the stabilisation of recombination intermediates by the ZMM proteins [77,78]. In Arabidopsis, SC polymerisation is also dependent on the formation of HR intermediates, as no synapsis is observed either in spo11, dmc1, or rad51 mutants where recombination is either not initiated or is blocked at the invasion step. However, Arabidopsis zmm mutants all display normal synapsis [20,48,69], showing that SC polymerisation in these species depends on the formation of

recombination intermediates, but not on their stabilisation by the ZMMs.

The limited synapsis progression observed in *axr1* mutants therefore suggests that recombination only proceeds far enough in a limited fraction of the genome where SC can polymerise and COs are formed. This appears to explain what is seen in Figure 8K, where mature HEI10 foci are concentrated on a few ZYP1 stretches. In that sense, aberrant SC formation illustrates that either recombination is blocked in a portion of the genome or that only a limited portion of the genome is competent to support recombination maturation, resulting in the loss of the obligatory CO. Nevertheless, the observation of nuclei where mature HEI10 foci are clustered on a single ZYP1 stretch, whereas the level of synapsis is high (as illustrated on Figure 8H) shows that the amount of ZYP1 polymerisation can be uncoupled from clustering of mature HEI10 foci. This suggests that CO clustering in *axr1* is not only a consequence of limited synapsis progression.

It is interesting to note that, within clusters of more than two class I CO foci, the distance among foci is not random, showing that they still display interference. During wild-type meiosis, no more than two adjacent foci could be scored, showing that these events are either less frequent or much more distant than in axr1. In this latter case, our results suggest that interference strength is considerably modified in axr1, resulting in CO clustering, at least in some areas of the genome. In Arabidopsis, interference strength is not uniform within chromosomes and increases toward the chromosome extremities [79]. This suggests that regional modification of interference parameters could be affected in axr1. In the future, it will therefore be crucial to determine whether CO clustering in *axr1* is region-specific or not. Regardless, the average number of final CO events is unchanged in *axr1*, suggesting that (i) the total number of class I COs is precisely controlled, (ii) this control is still active in *axr1*, and (iii) the mechanism underlying this CO homeostasis is independent of the obligatory CO mechanism.

### AXR1 Acts on Meiotic Recombination Through Activation of a CRL4 Complex

Neddylation stimulates several subclasses of cullin RING Ub ligases. We provide evidence that during meiotic recombination neddylation acts through cullin 4 activation to regulate the localisation of class I COs. Cullin 4 is a widely conserved cullin, involved in a large range of cellular and developmental controls, many of which are associated with genome integrity maintenance [80].

CRL4 complexes are composed of a CUL4 scaffold, a small RING domain containing RBX1 protein, a WD40-like repeatcontaining adaptor DDB1 (DNA-damage binding 1), and a substrate receptor subunit called DWD (DDB1-binding WD40 protein) or DCAF (DDB1- and CUL4-associated factor) [80]. Evidence for CRL4 functions in genome integrity control come from multiple sources and concern mostly cell responses to UV damage and replication controls by regulating the accumulation of the replication licensing factor CDT1. For example, DDB1- and/ or CUL4A-depleted human cells accumulate DSBs and have an activated ATM-ATR cell cycle checkpoint [81]. The budding yeast *cul8* mutants (cullin 8 is thought to be the functional homologue of cullin 4 in S. cerevisae) also accumulate DNA damage [82]. In fission yeast, mutation in Ddb1 increases the spontaneous mutation rate by more than 20-fold and prevents premeiotic S phase entry [83]. CRL4 activity is also required for the NER pathway by controlling the detection and processing of DNA lesions induced by UV in plants [41,84-86], but also in mammals, as loss of CUL4A in mice leads to an increase in susceptibility to UV skin cancer [87]. Evidence for the role of CRL4 complexes in DSB repair was also provided in *Drosophila*, where DDB1 depletion promotes loss of heterozygosity in somatic cells [88]. In addition, CRL4s complexes may also be involved in HR regulation, as in fission yeast, *ddb1* mutants are defective in HR probably by regulating the pool of available dNTPs [89].

Interestingly, we observed that hardly any COs are retained in double *axr1zmm* mutants, suggesting that the MUS81 recombination pathway may be shut down in *axr1*. Because this pathway accounts for only a small proportion of all COs in *Arabidopsis* [14,22], this disruption would not have a strong impact on meiosis. However, it could have a dramatic effect on somatic DNA repair, as the MUS81 pathway is one of the major pathways of somatic HR in eukaryotes [90]. It would therefore be interesting to study the involvement of AXR1 in somatic DNA recombination, above all considering that Dohmann and collaborators observed DSB accumulation in *axr1 (axr1-3* and *axr1-12)* somatic cells [91].

Considering the crucial role of CRL4s in genome maintenance and the activation of DNA repair pathways including HR, it is hardly surprising to find that it is involved in the regulation of meiotic HR in Arabidopsis. Considering the conservation of CRL4 functions across kingdoms, it is likely that the regulation of meiotic recombination by one (or several) CRL4 complex(es) will be also observed in other eukaryotes. Indeed, two converging studies in mice recently showed that cullin 4 is also required for meiosis also in mammals, as depletion of Cul4a (one of the two mammalian Cul4 genes) led to male infertility [42,43]. Whether this infertility is associated with early recombination [42] or later CO resolution defects [43] is still under debate. Nevertheless, the observation that MLH1 foci number is unchanged in *cul4a* but that a fraction of meiotic cells show pachytene bivalents without any MLH1 foci [43] is reminiscent of our data on axr1. Therefore, we propose that neddylation is acting on one or several CRL4 complex(es) to regulate the localisation of class I COs not only in Arabidopsis but also in mammals. In A. thaliana, there are more than 85 substrate receptor DWD domain proteins that can assemble with DDB1A or DDB1B or directly with CUL4-RBX1 to form CRL4 complexes [92,93]. Further studies will be necessary to identify which of these is acting during meiosis.

#### **Materials and Methods**

#### Plant Material

Ws-4 lines (including EGS344, EIC174, and EVM8) were obtained from the Versailles collection of *Arabidopsis* T-DNA transformants available at http://www-ijpb.versailles.inra.fr/en/sgap/equipes/variabilite/crg/[94].

Col-0 lines [including N877898 (Sail\_904E06) and N3076 = *axr1-12*] were obtained from the collection of T-DNA mutants from the Salk Institute Genomic Analysis Laboratory (Columbia accession) (SIGnAL, http://signal.salk.edu/cgi-bin/tdnaexpress) [95] and provided by NASC (http://nasc.nott.ac. uk/).

Other mutant alleles used in this study are as follows:  $msh4^{Ws}$  (EXY25) [48],  $msh5^{Col}$  (SALK\_026553) [96];  $hei10^{Ws}$  (EQO124) [48],  $zip4^{Col}$  (SALK\_068052) [69];  $mer3^{Col}$  (mer3-2, SALK\_091560) [20],  $mlh1^{Col}$  (SK\_25975) [48],  $mus81^{Col}$  (SALK\_107515) [14],  $rad51^{Col}$  (Gabi\_134A01) [97],  $mre11^{Col}$  (mre11-4, Salk\_067823),  $cul1-6^{Col}$  [60],  $axr6-2^{Col}$  (N3818) [61],  $axr6-3^{Col}$  (eta1) [62],  $cul3w^{Col}$  [63], and  $cul4-1^{Col}$  [64].

#### Growth Conditions

Plants were grown in a greenhouse (photoperiod 16 h/d and 8 h/night; temperature  $20^{\circ}$ C day and night; humidity 70%).

#### AXR1 Cloning

Screening for A. thaliana T-DNA (A. tumefaciens transferred DNA) insertions that provoke meiotic defects, we isolated three mutant lines: EGS344, EIC174, and EVM8. They all segregated 3:1 for reduced fertility, meiotic defects, and a bushy vegetative phenotype. Linkage analysis (as described by Grelon et al. [98]) showed that none of the mutations were linked with a T-DNA insertion. We therefore undertook a rough positional cloning of the three mutations as described by De Muyt et al. [99]. The most closely linked marker was chr1\_02991901 for all three mutants (based on 31 F2 mutant plants for EVM8, 31 for EGS344, and 31 for EIC174). Fine gene mapping was then carried out as described by De Muyt et al. [99] using chromosome 1 microsatellite markers located between 1,243,352 and 1,573,000 bp.

Among the predicted genes by TAIR10 SeqViewer server (http://www.arabidopsis.org/), we retained *AXR1* (At1G05180) as the best candidate, as *axr1* mutants were previously shown to display the same vegetative developmental defects as EGS344, EIC174, and EVM8 [44,46]. Sequencing of At1g05180 in the three mutant lines showed that all three are disrupted in this open reading frame (see below).

We further analysed the axr1 reference allele (axr1-12) and another insertion line (Sail\_904E06) available in the public databases (http://signal.salk.edu/). They all displayed the same meiotic phenotype as the previously isolated lines.

#### Molecular Characterisation of axr1 Alleles

Sequencing of At1g05180 in the EIC174 mutant line revealed a single nucleotide insertion in exon 6 (position 1364 of the genomic sequence, corresponding to nt 688 in the cDNA), leading to a premature stop codon (a 222 aa protein is produced instead of 540 aa in wild type). In the EGS344 mutant, a deletion of 898 bp (from nucleotide 91 of the genomic sequence) together with an insertion of *Agrobacterium* plasmid Ti DNA disrupts At1g05180 (Figure S1). In the EVM8 line, an in-frame deletion of 312 bp occurred between exons 3 and 4, generating a 20 aa deleted protein. Details are shown in Figure S1.

In *axr1-12*, corresponding to the N3076 line, a single C-T nucleotide substitution in position 1295 of the cDNA occurred, leading to a premature stop codon (415 aa instead of 540), as described by Leyser et al. [44]. In N877898, corresponding to the Sail\_904E06 line, a T-DNA insertion occurred in intron 11.

Sequence references are as follows: Tair Accession 4010763662 for the genomic sequence, and Tair Accession 4010730885 for the cDNA sequence.

#### PCR Genotyping of Mutant Lines

For EGS344 and EVM8, wild-type alleles were amplified with primers 05180-P1 (ACCCTGATTGAAGAAAAGTCT) and 05180-P2 (CGGAGGTCGTCAAGAAAA) (60°C, 30 PCR cycles, 1,200 bp). The EGS344 mutant allele was amplified with primers 05180-P1 and 05180-AgroP1 (ACATCACAGCACCTC-GATCCTGG) (60°C, 30 PCR cycles, 300 bp). The EVM8 mutant allele was amplified with 05180-P1 and 05180-P2 (60°C, 30 PCR cycles, 980 bp)

For N877898, the wild-type allele was amplified with primers N877898U and N877898L (60°C, 30 PCR cycles, 957 bp). The mutant allele was amplified with primers N877898L and Lb3SAIL (TAGCATCTGAATTTCATAACCAATCTCGATA-CAC) (60°C, 30 PCR cycles, 500 bp).

For all other genotypes, the primer list and PCR amplification conditions are shown in Table S6.

#### Genetic Analyses

**Recombination and interference measurements.** The six intervals used in this study correspond to intervals I5a, I5b, I5c, I5d I3b, and I3c, described by Berchowitz et al. [50] and in Table S3A.

We produced plants  $qrt^{-/-}$  N877898<sup>+/-</sup>, and  $qrt^{-/-}$ N877898<sup>+/-</sup> RYC/RYC. We crossed these two plants and in the progeny analysed tetrad fluorescence of semi-sterile plants  $art^{-/-}$  N877898<sup>-/-</sup> RYC/+++ or fertile plants either  $art^{-/-}$  $N877898^{+/-}$  RYC/+++ or  $art^{-/-}$  N877898<sup>+/+</sup> RYC/+++. Plants were grown in a greenhouse. Tetrad analyses were carried out as described in [50]. The resulting tetrad data (Table S3B) were analysed as described by Berchowitz et al. [50]. In brief, map distances were calculated using the Perkins mapping equation based on the measurement of the frequency of tetratype (T), parental (P), and NPD combinations of markers [d(cM) = (100)[6NPD+T])/(2[P+NPD+T])] [51]. Interference was then measured by comparing CO frequency in an interval when the adjacent interval had no CO to the CO frequency when the adjacent interval does have a CO, as described by Malkova et al. [18]. We calculated the ratio of these genetic distances and statistically compared these ratios as described by Berchowitz et al. [50] and using Stahl Lab Online tools (http://www.molbio. uoregon.edu/~fstahl/). Another estimate of interinterval interference via the coefficient of coincidence is shown in Table S4B. In addition, we estimated the level of intra-interval interference by calculating the NPD ratios (Table S5), which compares the number of observed double COs within a single interval (NPD) to the expected number of double COs under the hypothesis of no interference [52,100].

Interference analysis of *axr1* using the HEI10 and MLH1 foci patterns. Given a cluster of at least three MLH1 or HEI10 foci, we asked whether the internal foci were distributed as happens in the absence of interference. Because these clusters were small, under the hypothesis  $H_0$  of no interference, the foci should be distributed uniformly. Considering first the clusters formed with three foci, let  $d_1$  and  $d_2$  be the two distances between adjacent foci. After normalization, we have  $d_1+d_2=1$ . Then, we introduced the statistic *S* that corresponds to the sum of the squared centred deviations, normalized by the variance under  $H_0$ . Because the mean (respectively, the variance) of a uniformly distributed random variable in [0;1] is 0.5 (respectively, 1/12),  $S = [(d_1 - 0.5)^2 + (d_2 - 0.5)^2] * 12$ . This statistic can be generalized to clusters with k+2 foci (k+1 distances,  $d_1+d_2+...+d_{k+1}=1$ ):  $S = [(d_1-1/(k+1))^2+(d_2-1/(k+1))^2+...+(d_{k+1}-1/(k+1))^2] * (k+1)^2 * (k+2)/k$ .

The normalization is chosen so that under  $H_0$  each distance contributes to the same average to the statistic, regardless of the value of k. The total statistic S for the test is simply obtained by summing over all clusters. It is a random variable whose distribution we obtained by direct simulation under  $H_0$  of  $10^5$ datasets having the same values of k as in the experimental measurements. Small experimental values of S correspond to foci more regularly distributed than expected. The p value for our test is given by the proportion of simulated statistics smaller than that of the experimental data.

#### Antibodies

The anti-ASY1 polyclonal antibody was described by Armstrong et al. [101]. It was used at a dilution of 1:500. The anti-ZYP1 polyclonal antibody was described by Higgins et al. [57]. It was used at a dilution of 1:500. The anti-DMC1, anti-MLH1, and anti-HEI10 antibodies were described by Chelysheva et al. in [69], [49], and [48], respectively. These were used at a dilution of 1:20, 1:200, and 1:200, respectively. The anti-REC8 polyclonal antibody was described by Cromer et al. [102] and the anti-SCC3 by Chelysheva et al. [56]. These were used at a dilution of 1:250 and 1:500, respectively.

#### Microscopy

Comparison of the early stages of microsporogenesis and the development of PMCs was carried out as described in Grelon et al. [98]. Preparation of prophase stage spreads for immunocytology was performed using Carnoy's fixative and acetic acid chromosome spreads [59], except for DMC1 detection and double HEI10/ZYP1 immunolabelling where lipsol spreading and paraformaldehyde fixation were used [58]. Chiasma numbers were assessed by analysing metaphase I spread PMC chromosomes stained with DAPI, as described by Sanchez-Moran et al. [103]. In brief, a rod bivalent stands for a single chiasma, whereas a ring bivalent as two (one on each arm).

Observations were made as described by Chelysheva et al. [48].

#### **Supporting Information**

**Figure S1** Molecular characterisation of the *axr1* alleles. In EIC174, a single nucleotide (A) insertion occurred in exon 6 (position 1364 of the genomic sequence, in red), leading to a premature stop codon (a 222 aa protein is produced instead of 540 aa in wild type). In EVM8, a large in-frame deletion of 312 bp (in blue) generates a 20 aa truncated protein. In EGS344, a 907 bp deletion (in green) associated with an *Agrobacterium* Ti plasmid DNA insertion (\*) occurred in the 5' end of the gene. (DOCX)

**Figure S2** Phenotype of *axr1* mature plants. (A) *axr1* mutants display strong vegetative defects. All plants shown are 6 wk old. Upper panel, *axr1* allelic series. Lower panel, *axr1* mutants compared to their respective wild-type strain (Col-0 or Ws-4). (B) *axr1* mutants produce less seeds than wild type. (TIF)

**Figure S3** MLH1 foci localise on chiasma-containing arms in wild type and *axr1*. MLH1 was immunolocalised on acetic acid spread chromosomes from wild type (wt, Col-0) and *axr1* (N877898) at diakinesis. Because adjacent univalents cannot be distinguished from bivalents, we selected only the MLH1-labelled bivalent-like structures (arrows) and scored where MLH1 foci occurred. In both genotypes, 100% of the MLH1 foci (n = 246 for wild type, n = 44 for *axr1*) were detected on connected chromosome arms and never on a free chromosome. (TIF)

**Figure S4** Class I CO cluster characterisation. (A) The distance between two adjacent foci does not vary significantly with the cluster type. The distance between two adjacent foci (measured in  $\mu$ m and divided by the total size of the chromosome axis) was measured in clusters containing two, three, or four clustered foci. The *x* axis is the total number of foci in the considered clusters. All measures were undertaken in the N877898 allele. (B) Class I CO cluster lengths increase proportionally with the foci number. In the graph below, the mean size of the HEI10 or MLH1 clusters (calculated in  $\mu$ m and divided by the total size of the chromosome axis) is given according to the number of foci present in the cluster. All measures were undertaken in the N877898 allele. (TIF)

**Figure S5** Early recombination events are not altered in *axr1*. (A and B) DAPI staining of anaphase I meiocytes from rad51 (A) and *axr1rad51* (B) mutants. rad51 is defective in meiotic DSB repair as shown by the major chromosomal defects observed at meiosis

(fragmentation, A). Introgression of the *axr1* mutation (N877898 allele) does not rescue these defects (B), showing that meiotic DSB are present in *axr1*. Bars = 10  $\mu$ m. (C–F) Lipsol chromosome spreads of *Arabidopsis* meiocytes stained with DAPI (C and E) and immunolabelled with the anti-DMC1 antibody (D and F). *axr1* meiocytes (N877898 allele, E and F) show wild-type–like DMC1 staining. Bar = 5  $\mu$ m.

(TIF)

**Figure S6** No major axis defect can be detected in *axr1*. Immunolocalisation of ASY1 (A and B), REC8 (C and D), and SCC3 (E and F) in wild type (A, C, and E) and *axr1* (B, D, and F) prophase meiotic cells (N877898 allele). Bar = 5  $\mu$ m. (TIF)

**Figure S7** Synapsis is strongly perturbed in *axr1*. ZYP1 was immunolocalised on lipsol spread chromosomes from wild-type (A–D) and *axr1* (N877989 allele, E–L) meiotic cells. This figure corresponds to the red channel from Figure 8. (TIF)

**Figure S8** HEI10 dynamics during early prophase is unchanged in *axr1*. HEI10 was immunolocalised on lipsol spread chromosomes from wild-type (A–D) and *axr1* (N877989 allele, E–L) meiotic cells. This figure corresponds to the green channel from Figure 8.

(TIF)

 Table S1
 Average
 MCN
 and
 average
 bivalent
 number
 per

 meiocyte.
 Image: S1
 Image:

(DOCX)

**Table S2** Interfoci distance within class I CO clusters. In *axr1* (N877898 allele), HEI10 and MLH1 foci form clusters in approximately half of the pachytene/diplotene meiocytes. To estimate interference between adjacent foci within the clusters, we measured the distance between two adjacent foci in clusters containing more than two foci.

(DOCX)

**Table S3** Recombination dataset. (A) The localisation of the six
 FTL intervals used in this study are shown below on the five Arabidopsis chromosomes. They correspond to three pairs of linked intervals-I5a I5b, I5c I5d, and I3b I3c-described by Berchowitz and Copenhaver [50]. The position (in bp) of each transgene encoding red, yellow, or cyan fluorescent proteins (filled circles) is shown. The size of each interval is given in the accompanying table. The map was obtained using the Chromosome Map Tool from http://www.arabidopsis.org/jsp/ ChromosomeMap/tool.jsp. (B) To measure recombination rates, we produced plants homozygous for the quartet (qrt) mutation, heterozygous for axr1 (N877898 allele), and carrying or not three linked fluorescent markers (R, red; Y, yellow; C, cyan): qrt $axr1^{+/-}$  and  $qrt^{-/-}axr1^{+/-}$  RYC/RYC. The qrt mutation allows the four pollen grains from a single meiosis to be maintained together. We crossed these two plants and in the progeny analysed tetrad fluorescence of mutant plants  $(qrt^{-/-} axrI^{-/-} RYC/+++)$ or fertile plants (wt) (either  $qrt^{-/-} axrI^{+/-} RYC/+++$  or  $qrt^{-/-}$ axr1<sup>+/+</sup> RYC/+++). Plants were grown in a greenhouse, and tetrad analyses were carried out as described by Berchowitz and colleagues [50]. The distribution of the markers within the tetrads and the resulting distribution of colours vary depending on the number, localisation, and chromatids involved in recombination. The different possibilities are indicated by drawings and the number of tetrads in each phenotypic class is shown in the tables below. Two replicates were made for the I5a I5b intervals. (DOCX)

Tables S4 Interinterval interference analyses. Genetic interference among COs occurring in two linked intervals was calculated using two methods: (A) The IR. The IR [18] compares the genetic size of an interval (d(Ia), in cM calculated using the Perkins equation [51]) when a CO occurs in an adjacent interval (d(Ia) with CO in Ib) to the genetic size of the same interval when no CO occurs in the adjacent interval (d(Ia) without CO in Ib). The ratio of these two distances, called the IR, gives a measurement of the strength of interference between the two intervals. When there is no interference, the ratio is equal to 1, whereas the ratio is below 1 when there is interference. Ratios above 1 indicate negative interference, indicative of more adjacent C0s than expected. We compared each IR to 1 [P(IR = 1)] and compared mutant IR to wild-type IR [P(IR = IR wt)]. Calculations and statistical analyses were performed according to Berchowitz and Copenhaver [50] using Stahl Lab Online tools (http://www.molbio.uoregon.edu/ ~fstahl/). In wild type, IRs were statistically below 1 for all pairs of intervals considered, indicative of interference among COs. In axr1, IRs increased systematically and were always greater than 1, but the difference with 1 was significant (p < 0.01) only for intervals I5a I5b and only for one replicate of the experiment. (B) The coefficient of coincidence. The coefficient of coincidence (c.o.c.) compares the observed frequency of double COs in the two adjacent intervals [f(Ia and Ib) observed] to the expected frequency of double COs if there were no interference [f(Ia and Ib) expected]. This last frequency is the product of the frequency of COs in each single interval [f(Ia) and f(Ib)]. The c.o.c. corresponds to (f observed/f expected). When interference is absent, the c.o.c. is equal to 1. When interference is total, the c.o.c. is equal to 0. For wild type, this index varied from 0.37 to 0.63, indicative of interference. In axr1 the index was always above 1, which shows that double COs in adjacent intervals are more frequent than expected. (DOCX)

**Table S5** Intra-interval interference analyses. Interference was

#### References

1. Jones GH (1984) The control of chiasma distribution. Symp Soc Exp Biol 38: 293–320.

measured within a single interval, by comparing the observed

- Jones GH, Franklin FCH (2006) Meiotic crossing-over: obligation and interference. Cell 126: 246–248. doi:10.1016/j.cell.2006.07.010.
- Gerton JL, Hawley RS (2005) Homologous chromosome interactions in meiosis: diversity amidst conservation. Nat Rev Genet 6: 477–487. Available: http://www.ncbi.nlm.nih.gov/pubmed/15931171. Accessed 9 November 2012.
- Mézard C, Vignard J, Drouaud J, Mercier R (2007) The road to crossovers: plants have their say. Trends Genet 23: 91–99. Available: http://www.ncbi. nlm.nih.gov/pubmed/17208327. Accessed 19 November 2012.
- Keeney S, Neale MJ (2006) Initiation of meiotic recombination by formation of DNA double-strand breaks: mechanism and regulation. Biochem Soc Trans 34: 523–525. Available: http://www.ncbi.nlm.nih.gov/pubmed/16856850.
- Hunter N, Kleckner N (2001) The single-end invasion: an asymmetric intermediate at the double-strand break to double-Holliday junction transition of meiotic recombination. Cell 106: 59–70. doi:10.1016/S0092-8674/01)00430-5.
- Kagawa W, Kurumizaka H (2010) From meiosis to postmeiotic events: uncovering the molecular roles of the meiosis-specific recombinase Dmc1. FEBS J 277: 590–598. doi:10.1111/j.1742-4658.2009.07503.x.
- Schwacha a, Kleckner N (1995) Identification of double Holliday junctions as intermediates in meiotic recombination. Cell 83: 783–791. Available: http:// www.ncbi.nlm.nih.gov/pubmed/8521495.
- Bishop DK, Zickler D (2004) Early decision: meiotic crossover interference prior to stable strand exchange and synapsis review. Cell 117: 9–15.
- Youds JL, Boulton SJ (2011) The choice in meiosis defining the factors that influence crossover or non-crossover formation. J Cell Sci 124: 501–513. Available: http://www.ncbi.nlm.nih.gov/pubmed/21282472. Accessed 8 November 2012.
- Pâques F, Haber JE (1999) Multiple pathways of recombination induced by double-strand breaks in Saccharomyces cerevisiae. Microbiol Mol Biol Rev 63:

number of double COs [based on NPD frequency (NPD observed) to the expected number of double COs under the hypothesis of no interference (NPD expected)] [52]. The ratio between these two figures (NPDr) gives the strength of interference within the considered interval. The NPD tetrads correspond to h, i, j, k, and l classes from Table S2. In wild type, NPDr indicate strong interference (NPDr close to 0.3) within all the intervals (except for I3c, which is too small to give statistically meaningful figures). In *axr1*, NPDrs increased systematically (between 0.7 and 2.69) and were generally greater than 1, indicative of a trend toward negative interference (more double COs in a single interval than expected). However, statistical analyses on NPDr (Stahl Lab Online tools, http://www.molbio.uoregon.edu/~fstahl/) showed that these values are statistically different from 1 (p<0.01) only on I5a and I5b (one of the two replicates).

(DOCX)

**Table S6**Primer sequences and PCR conditions for mutantgenotyping.

(DOCX)

#### Acknowledgments

We are grateful to Wayne Crismani, Eric Jenczewski, Christine Mézard, Raphaël Mercier, and Wojtek Pawlowski for helpful discussions and constructive reading of the manuscript. We wish to thank Gregory Copenhaver for providing FTL lines and Pascal Genschik for providing *cul3w* and *cul4-1* lines. We also wish to thank LKG Scientific Editing & Translation for correcting the manuscript.

#### **Author Contributions**

The author(s) have made the following declarations about their contributions: Conceived and designed the experiments: MTJ MG. Performed the experiments: MTJ DV AC LP LC. Analyzed the data: MTJ DV AC LP MF OCM LC MG. Wrote the paper: MTJ MF OCM MG.

349–404. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 98970&tool = pmcentrez&rendertype = abstract.

- Sturtevant AH (1915) The behavior of the chromosomes as studied through linkage. New York: Columbia University.
- Muller HJ (1916) The mechanism of crossing-over. New York: Columbia University.
- 14. Berchowitz LE, Francis KE, Bey AL, Copenhaver GP (2007) The role of AtMUS81 in interference-insensitive crossovers in A. thaliana. PLoS Genet 3: e132. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 1941751&tool = pmcentrez&rendertype = abstract. Accessed 19 November 2012.
- Copenhaver GP, Housworth EA, Stahl FW (2002) Crossover interference in Arabidopsis. Genetics 160: 1631–1639.
- De los Santos T, Hunter N, Lee C, Larkin B, Loidl J, et al. (2003) The Mus81/ Mms4 endonuclease acts independently of double-Holliday junction resolution to promote a distinct subset of crossovers during meiosis in budding yeast. Genetics 164: 81–94.
- Housworth EA, Stahl FW (2003) Crossover interference in humans. Am J Hum Genet 73: 188–197. doi:10.1086/376610.
- Malkova A, Swanson J, German M, McCusker JH, Housworth EA, et al. (2004) Gene conversion and crossing over along the 405-kb left arm of Saccharomyces cerevisiae chromosome VII. Genetics 168: 49–63. doi:10.1534/genetics.104.027961.
- Higgins JD, Armstrong SJ, Franklin FCH, Jones GH (2004) The Arabidopsis MutS homolog AtMSH4 functions at an early step in recombination: evidence for two classes of recombination in Arabidopsis. Genes Dev 18(20): 2557–2570. doi:10.1101/gad.317504.eukaryote.
- Mercier R, Jolivet S, Vezon D, Huppe E, Chelysheva L, et al. (2005) Two meiotic crossover classes cohabit in Arabidopsis: one is dependent on MER3,whereas the other one is not. Curr Biol 15: 692–701. Available: http://www.ncbi.nlm.nih.gov/pubmed/15854901. Accessed 4 June 2013.
- Osman KE, Higgins JD, Sanchez-Moran E, Armstrong SJ, Franklin FCH (2011) Pathways to meiotic recombination in Arabidopsis thaliana. New Phytol

- Higgins JD, Buckling EF, Franklin FCH, Jones GH (2008) Expression and functional analysis of AtMUS81 in Arabidopsis meiosis reveals a role in the second pathway of crossing-over. Plant J 54: 152–162. Available: http://www. ncbi.nlm.nih.gov/pubmed/18182028. Accessed 19 November 2012.
- Berchowitz LE, Copenhaver GP (2010) Genetic interference: don't stand so close to me. Curr Genomics 11: 91–102. Available: http://www. pubmedcentral.nih.gov/articlerender.fcgi?artid = 2874225&tool = pmcentrez& rendertype = abstract.
- Henderson IR (2012) Control of meiotic recombination frequency in plant genomes. Curr Opin Plant Biol 15: 556–561. Available: http://www.ncbi.nlm. nih.gov/pubmed/23017241. Accessed 21 October 2013.
- Smalle J, Vierstra R (2004) The ubiquitin 26S proteasome proteolytic pathway. Annu Rev Plant Biol 55: 555–590.
- Duda DM, Scott DC, Calabrese MF, Zimmerman ES, Zheng N, et al. (2011) Structural regulation of cullin-RING ubiquitin ligase complexes. Curr Opin Struct Biol 21: 257–264. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 3151539&tool = pmcentrez&rendertype = abstract. Accessed 29 October 2013.
- Saha A, Deshaies RJ (2008) Multimodal activation of the ubiquitin ligase SCF by Nedd8 conjugation. Mol Cell 32: 21–31.
- Lyapina S, Cope G, Shevchenko A, Serino G, Tsuge T, et al. (2001) Promotion of NEDD-CUL1 conjugate cleavage by COP9 signalosome. Science 292: 1382–1385. doi:10.1126/science.1059780.
- Tateishi K, Omata M, Tanaka K, Chiba T (2001) The NEDD8 system is essential for cell cycle progression and morphogenetic pathway in mice. J Cell Biol 155: 571–579. doi:10.1083/jcb.200104035.
- Lammer D, Mathias N, Laplaza JM, Jiang W, Liu Y, et al. (1998) Modification of yeast Cdc53p by the ubiquitin-related protein Rub1p affects function of the SCFCdc4 complex. Genes Dev 12: 914–926. doi:10.1101/gad.12.7.914.
- Kurz T, Pintard L, Willis JH, Hamill DR, Gönczy P, et al. (2002) Cytoskeletal regulation by the Nedd8 ubiquitin-like protein modification pathway. Science 295: 1294–1298. doi:10.1126/science.1067765.
- Pintard L, Kurz T, Glaser S, Willis JH, Peter M, et al. (2003) Neddylation and deneddylation of CUL-3 is required to target MEI-1/Katanin for degradation at the meiosis-to-mitosis transition in C. elegans. Curr Biol 13: 911–921. doi:10.1016/S0960-9822(03)00336-1.
- 33. Santner A, Estelle M (2010) The ubiquitin-proteasome system regulates plant hormone signaling. Plant J 61: 1029–1040. Available: http://www. pubmedcentral.nih.gov/articlerender.fcgi?artid = 3066055&tool = pmcentrez& rendertype = abstract. Accessed 29 August 2013.
- Dreher K, Callis J (2007) Ubiquitin, hormones and biotic stress in plants. Ann Bot 99: 787–822. doi:10.1093/aob/mcl255.
- 35. Quint M, Gray WM (2006) Auxin signaling. Curr Opin Plant Biol 9: 448-453.
- Mockaitis K, Éstelle M (2008) Auxin receptors and plant development: a new signaling paradigm. Annu Rev Cell Dev Biol 24: 55–80. doi:10.1146/ annurev.cellbio.23.090506.123214.
- Hua Z, Vierstra RD (2011) The cullin-RING ubiquitin-protein ligases. Annu Rev Plant Biol 62: 299–334. Available: http://www.ncbi.nlm.nih.gov/ pubmed/21370976. Accessed 29 October 2013.
- Pozo J Del, Estelle M (1999) The Arabidopsis cullin AtCUL1 is modified by the ubiquitin-related protein RUB1. Natl Acad Sci U S A 96: 15342–15347.
- El-Mahdy M a, Zhu Q, Wang Q, Wani G, Praetorius-Ibba M, et al. (2006) Cullin 4A-mediated proteolysis of DDB2 protein at DNA damage sites regulates in vivo lesion recognition by XPC. J Biol Chem 281: 13404–13411. Available: http://www.ncbi.nlm.nih.gov/pubmed/16527807. Accessed 28 October 2013.
- Fischer ES, Scrima A, Böhm K, Matsumoto S, Lingaraju GM, et al. (2011) The molecular basis of CRL4DDB2/CSA ubiquitin ligase architecture, targeting, and activation. Cell 147: 1024–1039. Available: http://www.ncbi.nlm.nih. gov/pubmed/22118460. Accessed 23 October 2013.
- 41. Molinier J, Lechner E, Dumbliauskas E, Genschik P (2008) Regulation and role of Arabidopsis CUL4-DDB1A-DDB2 in maintaining genome integrity upon UV stress. PLoS Genet 4: e1000093. Available: http://www.pubmedcentral.nih. gov/articlerender.fcgi?artid = 2396500&tool = pmcentrez&rendertype = abstract. Accessed 8 August 2013.
- 42. Kopanja D, Roy N, Stoyanova T, Hess R a, Bagchi S, et al. (2011) Cul4A is essential for spermatogenesis and male fertility. Dev Biol 352: 278– 287. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid=3065526&tool=pmcentrez&rendertype=abstract. Accessed 19 November 2012.
- 43. Yin Y, Lin C, Kim ST, Roig I, Chen H, et al. (2011) The E3 ubiquitin ligase Cullin 4A regulates meiotic progression in mouse spermatogenesis. Dev Biol 356: 51–62. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 3130830&tool = pmcentrez&rendertype = abstract. Accessed 31 October 2012.
- 44. Leyser HM, Lincoln CA, Timpte C, Lammer D, Turner J, et al. (1993) Arabidopsis auxin-resistance gene AXR1 encodes a protein related to ubiquitin-activating enzyme E1. Nature 364: 161–164. Available: http:// www.ncbi.nlm.nih.gov/pubmed/8321287. Accessed 29 October 2013.
- Estelle MA, Somerville C (1987) Auxin-resistant mutants of Arabidopsis thaliana with an altered morphology. MGG Mol Gen Genet 206: 200–206. doi:10.1007/BF00333575.

- Lincoln C, Britton JH, Estelle M (1990) Growth and development of the axr1 mutants of Arabidopsis. Plant Cell 2: 1071-1080. doi:10.1105/tpc.2.11.1071.
- Zickler D, Kleckner N (1999) Meiotic chromosomes: integrating structure and function. Annu Rev Genet 33: 603–754. Available: http://www.ncbi.nlm.nih. gov/pubmed/10690419. Accessed 22 August 2013.
- 48. Chelysheva L, Vezon D, Chambon A, Gendrot G, Pereira L, et al. (2012) The Arabidopsis HEI10 is a new ZMM protein related to Zip3. PLoS Genet 8: e1002799. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 3405992&tool = pmcentrez&rendertype = abstract. Accessed 10 July 2013.
- 49. Chelysheva L, Grandont L, Vrielynck N, le Guin S, Mercier R, et al. (2010) An easy protocol for studying chromatin and recombination protein dynamics during Arabidopsis thaliana meiosis: immunodetection of cohesins, histones and MLH1. Cytogenet Genome Res 129: 143–153. Available: http://www. ncbi.nlm.nih.gov/pubmed/20628250. Accessed 10 July 2013.
- Berchowitz LE, Copenhaver GP (2008) Fluorescent Arabidopsis tetrads: a visual assay for quickly developing large crossover and crossover interference data sets. Nat Protoc 3: 41–50. Available: http://www.ncbi.nlm.nih.gov/ pubmed/18193020. Accessed 19 November 2012.
- Perkins DD (1949) Biochemical mutants in the smut fungus ustilago maydis. Genetics 34: 607–625.
- 52. Stahl FW (2008) On the "NPD ratio" as a test for crossover interference. Genetics 179: 701–704. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 2390647&tool = pmcentrez&rendertype = abstract. Accessed 19 November 2012.
- Storlazzi A, Tesse S, Ruprich-Robert G, Gargano S, Pöggeler S, et al. (2008) Coupling meiotic chromosome axis integrity to recombination. Genes Dev 22: 796–809. doi:10.1101/gad.459308.
- Kleckner N (2006) Chiasma formation: chromatin/axis interplay and the role(s) of the synaptonemal complex. Chromosoma 115: 175–194. Available: http:// www.ncbi.nlm.nih.gov/pubmed/16555016. Accessed 7 March 2013.
- Cai X, Dong F, Edelmann RE, Makaroff CA (2003) The Arabidopsis SYN1 cohesin protein is required for sister chromatid arm cohesion and homologous chromosome pairing. J Cell Sci 116: 2999–3007. Available: http://www.ncbi. nlm.nih.gov/pubmed/12783989. Accessed 5 July 2013.
- Chelysheva L, Diallo S, Vezon D, Gendrot G, Vrielynck N, et al. (2005) AtREC8 and AtSCC3 are essential to the monopolar orientation of the kinetochores during meiosis. J Cell Sci 118: 4621–4632. Available: http:// www.ncbi.nlm.nih.gov/pubmed/16176934. Accessed 10 July 2013.
- Higgins JD, Sanchez-moran E, Armstrong SJ, Jones GH, Franklin FCH (2005) The Arabidopsis synaptonemal complex protein ZYP1 is required for chromosome synapsis and normal fidelity of crossing over. Genes Dev 19(20): 2488–2500. doi:10.1101/gad.354705.ments.
- Armstrong S, Osman KE (2013) Immunolocalization of meiotic proteins in Arabidopsis thaliana: method 2. Methods Mol Biol 990: 103–107. Available: http://www.ncbi.nlm.nih.gov/pubmed/23559206. Accessed 21 March 2014.
- Chelysheva LA, Grandont L, Grelon M (2013) Immunolocalization of meiotic proteins in Brassicaceae: method 1. In: Springer Protocols, editor. Methods in molecular biology (Clifton, N.J.). Humana Press, Vol. 990. pp. 93–101. Available: http://www.ncbi.nlm.nih.gov/pubmed/23559205. Accessed 11 July 2013.
- Moon J, Zhao Y, Dai X, Zhang W, Gray WM, et al. (2007) A new CULLIN 1 mutant has altered responses to hormones and light in Arabidopsis. Plant Physiol 143: 684–696. doi:10.1104/pp.106.091439.
- Ren C, Pan J, Peng W, Genschik P, Hobbie L, et al. (2005) Point mutations in Arabidopsis Cullin1 reveal its essential role in jasmonate response. Plant J 42: 514–524. doi:10.1111/j.1365-313X.2005.02394.x.
- Quint M, Ito H, Zhang W, Gray WM (2005) Characterization of a novel temperature-sensitive allele of the CUL1/AXR6 subunit of SCF ubiquitinligases. Plant J 43: 371–383. doi:10.1111/j.1365-313X.2005.02449.x.
- 63. Thomann A, Lechner E, Hansen M, Dumbliauskas E, Parmentier Y, et al. (2009) Arabidopsis CULLIN3 genes regulate primary root growth and patterning by ethylene-dependent and -independent mechanisms. PLoS Genet 5: e1000328. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 2607017&tool = pmcentrez&rendertype = abstract. Accessed 5 September 2013.
- 64. Bernhardt A, Lechner E, Hano P, Schade V, Dieterle M, et al. (2006) CUL4 associates with DDB1 and DET1 and its downregulation affects diverse aspects of development in Arabidopsis thaliana. Plant J 47: 591–603. Available: http://www.ncbi.nlm.nih.gov/pubmed/16792691. Accessed 3 September 2013.
- 65. Zhang L, Liang Z, Hutchinson J, Kleckner N (2014) Crossover patterning by the beam-film model: analysis and implications. PLoS Genet 10: e1004042. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid=3907302&tool=pmcentrez&rendertype=abstract. Accessed 9 April 2014.
- Fung JC, Rockmill B, Odell M, Roeder GS (2004) Imposition of crossover interference through the nonrandom distribution of synapsis initiation complexes. Cell 116: 795–802. Available: http://www.ncbi.nlm.nih.gov/ pubmed/15035982.
- Crismani W, Girard C, Froger N, Pradillo M, Santos JL, et al. (2012) FANCM limits meiotic crossovers. Science 336: 1588–1590. Available: http://www. ncbi.nlm.nih.gov/pubmed/22723424. Accessed 31 October 2012.

- 68. Getz TJ, Banse S a, Young LS, Banse A V, Swanson J, et al. (2008) Reduced mismatch repair of heteroduplexes reveals "non"-interfering crossing over in wild-type Saccharomyces cerevisiae. Genetics 178: 1251–1269. Available: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid = 2278109&tool = pmcentrez&rendertype = abstract. Accessed 13 December 2013.
- 69. Chelysheva L, Gendrot G, Vezon D, Doutriaux M-P, Mercier R, et al. (2007) Zip4/Spo22 is required for class I CO formation but not for synapsis completion in Arabidopsis thaliana. PLoS Genet 3: e83. Available: http:// www.pubmedcentral.nih.gov/articlerender.fcgi?artid = 1877879&tool = pmcentrez& rendertype = abstract. Accessed 10 July 2013.
- 70. Joshi N, Barot A, Jamison C, Börner GV (2009) Pch2 links chromosome axis remodeling at future crossover sites and crossover distribution during yeast meiosis. PLoS Genet 5: e1000557. Available: http://www.pubmedcentral.nih. gov/articlerender.fcgi?artid = 2708914&tool = pmcentrez&rendertype = abstract. Accessed 1 November 2012.
- 71. Zanders S, Alani E (2009) The pch2Delta mutation in baker's yeast alters meiotic crossover levels and confers a defect in crossover interference. PLoS Genet 5: e1000571. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 2709914&tool = pmcentrez&rendertype = abstract. Accessed 1 November 2012.
- Zickler D (2006) From early homologue recognition to synaptonemal complex formation. Chromosoma 115: 158–174. Available: http://www.ncbi.nlm.nih. gov/pubmed/16570189. Accessed 13 December 2013.
- Shinohara M, Oh SD, Hunter N, Shinohara A (2008) Crossover assurance and crossover interference are distinctly regulated by the ZMM proteins during yeast meiosis. Nat Genet 40: 299–309. Available: http://www.ncbi.nlm.nih. gov/pubmed/18297071. Accessed 19 November 2012.
- Bishop DK, Zickler D (2004) Early decision; meiotic crossover interference prior to stable strand exchange and synapsis. Cell 117: 9–15. Available: http:// www.ncbi.nlm.nih.gov/pubmed/15066278. Accessed 15 April 2014.
- Libuda DE, Uzawa S, Meyer BJ, Villeneuve AM (2013) Meiotic chromosome structures constrain and respond to designation of crossover sites. Nature 502(7473): 703–706. Available: http://www.nature.com/doifinder/10.1038/ nature12577. Accessed 9 October 2013.
- 76. Wang M, Wang K, Tang D, Wei C, Li M, et al. (2010) The central element protein ZEP1 of the synaptonemal complex regulates the number of crossovers during meiosis in rice. Plant Cell 22: 417–430. Available: http://www. pubmedcentral.nih.gov/articlerender.fcgi?artid = 2845403&tool = pmcentrez& rendertype = abstract. Accessed 19 November 2012.
- Lynn A, Soucek R, Börner GV (2007) ZMM proteins during meiosis: crossover artists at work. Chromosome Res 15: 591–605. Available: http://www.ncbi. nlm.nih.gov/pubmed/17674148. Accessed 26 October 2012.
- Börner GV, Kleckner N, Hunter N (2004) Crossover/noncrossover differentiation, synaptonemal complex formation, and regulatory surveillance at the leptotene/zygotene transition of meiosis. Cell 117: 29–45. Available: http:// www.ncbi.nlm.nih.gov/pubmed/15066280.
- Basu-Roy S, Gauthier F, Giraut L, Mézard C, Falque M, et al. (2013) Hot regions of noninterfering crossovers coexist with a nonuniformly interfering pathway in Arabidopsis thaliana. Genetics 195: 769–779. Available: http:// www.ncbi.nlm.nih.gov/pubmed/24026099. Accessed 10 April 2014.
- Jackson S, Xiong Y (2009) CRL4s: the CUL4-RING E3 ubiquitin ligases. Trends Biochem Sci 34: 562–570. Available: http://www.pubmedcentral.nih. gov/articlerender.fcgi?artid = 2783741&tool = pmcentrez&rendertype = abstract. Accessed 4 November 2013.
- Lovejoy C a, Lock K, Yenamandra A, Cortez D (2006) DDB1 maintains genome integrity through regulation of Cdt1. Mol Cell Biol 26: 7977–7990. Available: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=1636754& tool=pmcentrez&rendertype=abstract. Accessed 13 December 2013.
- Luke B, Versini G, Jaquenoud M, Zaidi IW, Kurz T, et al. (2006) The cullin Rtt101p promotes replication fork progression through damaged DNA and natural pause sites. Curr Biol 16: 786–792. Available: http://www.ncbi.nlm. nih.gov/pubmed/16631586. Accessed 13 December 2013.
- Holmberg C, Fleck O, Hansen HA, Liu C, Slaaby R, et al. (2005) Ddb1 controls genome stability and meiosis in fission yeast. Genes Dev 19: 853–862. doi:10.1101/gad.329905.
- 84. Castells E, Molinier J, Benvenuto G, Bourbousse C, Zabulon G, et al. (2011) The conserved factor DE-ETIOLATED 1 cooperates with CUL4-DDB1DDB2 to maintain genome integrity upon UV stress. EMBO J 30: 1162–1172. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 3061027&tool = pmcentrez&rendertype = abstract. Accessed 4 November 2013.
- Biedermann S, Hellmann H (2011) WD40 and CUL4-based E3 ligases: lubricating all aspects of life. Trends Plant Sci 16: 38–46. Available: http:// www.ncbi.nlm.nih.gov/pubmed/20965772. Accessed 1 November 2012.

- Zhang C, Guo H, Zhang J, Guo G, Schumaker KS, et al. (2010) Arabidopsis cockayne syndrome A-like proteins 1A and 1B form a complex with CULLIN4 and damage DNA binding protein 1A and regulate the response to UV irradiation. Plant Cell 22: 2353–2369. doi:10.1105/tpc.110.073973.
- Liu L, Lee S, Zhang J, Peters SB, Hannah J, et al. (2009) CUL4A abrogation augments DNA damage response and protection against skin carcinogenesis. Mol Cell 34: 451–460. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 2722740&tool = pmcentrez&rendertype = abstract. Accessed 7 November 2013.
- Shimanouchi K, Takata K-I, Yamaguchi M, Murakami S, Ishikawa G, et al. (2006) Drosophila damaged DNA binding protein 1 contributes to genome stability in somatic cells. J Biochem 139: 51–58. doi:10.1093/jb/mvj006.
- 89. Moss J, Tinline-Purvis H, Walker C a, Folkes LK, Stratford MR, et al. (2010) Break-induced ATR and Ddb1-Cul4(Cdt)<sup>2</sup> ubiquitin ligase-dependent nucleotide synthesis promotes homologous recombination repair in fission yeast. Genes Dev 24: 2705–2716. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid = 2994043&tool = pmcentrez&rendertype = abstract. Accessed 7 November 2013.
- Heyer W-D, Ehmsen KT, Liu J (2010) Regulation of homologous recombination in eukaryotes. Annu Rev Genet 44: 113–139. doi:10.1146/ annurev-genet-051710-150955.
- Dohmann EMN, Levesque MP, De Veylder L, Reichardt I, Jürgens G, et al. (2008) The Arabidopsis COP9 signalosome is essential for G2 phase progression and genomic stability. Development 135: 2013–2022. Available: http://www.ncbi.nlm.nih.gov/pubmed/18434413. Accessed 4 November 2013.
- Guo L, Nezames CD, Sheng L, Deng X, Wei N (2013) Cullin-RING ubiquitin ligase family in plant abiotic stress pathways(F). J Integr Plant Biol 55: 21–30. doi:10.1111/jipb.12019.
- 93. Lee J-H, Terzaghi W, Gusmaroli G, Charron J-BF, Yoon H-J, et al. (2008) Characterization of Arabidopsis and rice DWD proteins and their roles as substrate receptors for CUL4-RING E3 ubiquitin ligases. Plant Cell 20: 152– 167. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 2254929&tool = pmcentrez&rendertype = abstract. Accessed 23 June 2014.
- Bechtold N, Ellis J, Pelletier G (1993) In Planta, Agrobacterium mediated gene transfer by integration of adult Arabidopsis thaliana plants. C R Acad Sci Paris 316: 1194–1199.
- Alonso JM, Stepanova AN, Leisse TJ, Kim CJ, Chen H, et al. (2003) Genomewide insertional mutagenesis of Arabidopsis thaliana. Science 301: 653–657. Available: http://www.ncbi.nlm.nih.gov/pubmed/12893945. Accessed 21 November 2013.
- 96. Higgins JD, Vignard J, Mercier R, Pugh AG, Franklin FCH, et al. (2008) AtMSH5 partners AtMSH4 in the class I meiotic crossover pathway in Arabidopsis thaliana, but is not required for synapsis. Plant J 55: 28–39. Available: http://www.ncbi.nlm.nih.gov/pubmed/18318687. Accessed 19 November 2012.
- 97. Li W, Chen C, Markmann-Mulisch U, Timofejeva L, Schmelzer E, et al. (2004) The Arabidopsis AtRAD51 gene is dispensable for vegetative development but required for meiosis. Proc Natl Acad Sci U S A 101: 10596–10601. Available: http://www.pubmedcentral.nih.gov/articlerender. fcgi?artid = 489980&tool = pmcentre&rendertype = abstract.
- Grelon M, Vezon D, Gendrot G, Pelletier G (2001) AtSPO11-1 is necessary for efficient meiotic recombination in plants. EMBO J 20: 589–600. Available: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid = 133473&tool = pmcentrez&rendertype = abstract. Accessed 10 July 2013.
- 99. De Muyt A, Pereira L, Vezon D, Chelysheva L, Gendrot G, et al. (2009) A high throughput genetic screen identifies new early meiotic recombination functions in Arabidopsis thaliana. PLoS Genet 5: e1000654. Available: http://www. pubmedcentral.nih.gov/articlerender.fcgi?artid = 2735182&tool = pmcentrez& rendertype = abstract. Accessed 10 July 2013.
- 100. Papazian HP. (1952) The analysis of tetrad data. Genetics 37: 175-188.
- 101. Armstrong SJ (2002) Asy1, a protein required for meiotic chromosome synapsis, localizes to axis-associated chromatin in Arabidopsis and Brassica. J Cell Sci 115: 3645–3655. Available: http://jcs.biologists.org/cgi/doi/10. 1242/jcs.00048. Accessed 19 November 2012.
- 102. Cromer L, Jolivet S, Horlow C, Chelysheva L, Heyman J, et al. (2013) Centromeric cohesion is protected twice at meiosis, by SHUGOSHINs at anaphase I and by PATRONUS at interkinesis. Curr Biol 23: 2090–2099. Available: http://www.ncbi.nlm.nih.gov/pubmed/24206843. Accessed 12 December 2013.
- Sanchez-Moran E, Armstrong SJ, Santos JL, Franklin FC, Jones GH (2001) Chiasma formation in Arabidopsis thaliana accession Wassileskija and in two meiotic mutants. Chromosome Res 9: 121–128. Available: http://www.ncbi. nlm.nih.gov/pubmed/11321367.